Author’s response to reviews

Title: Cyclosporine-A Attenuates Retinal Inflammation by Inhibiting HMGB-1 Formation in Rats with Type 2 Diabetes Mellitus

Authors:

Peng Wang (luckywp2000@aliyun.com)
Fei Chen (731050214@qq.com)
Xuedong Zhang (zxued@sina.com)

Version: 2 Date: 16 Jan 2020

Author's response to reviews:

According to the editor's comments, we have comprehensively and thoroughly examined the Method part and the Discussion part, and re-phrased these sections to minimise overlap.

Thanks very much for your comments!

We really appreciate your work for the review!